Author's response to reviews

Title: Effectiveness of computer-navigated minimally invasive total hip surgery compared to conventional total hip arthroplasty: design of a randomized controlled trial

Authors:

Inge H.F. Reininga (i.reininga@orth.umcg.nl)
Robert Wagenmakers (r.wagenmakers@orth.umcg.nl)
Inge van den Akker-Scheek (i.scheek@orth.umcg.nl)
A. Dennis Stant (d.stant@mta.umcg.nl)
Johan W. Groothoff (j.w.groothoff@med.umcg.nl)
Sjoerd K. Bulstra (s.j.bulstra@orth.umcg.nl)
Wiebren Zijlstra (w.zijlstra@med.umcg.nl)
Martin Stevens (m.stevens@orth.umcg.nl)

Version: 2 Date: 10 January 2007

Author's response to reviews: see over
To:
Editor of BMC Musculoskeletal Disorders
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Date: 10-01-2007
Subject: Submission revised manuscript

Dear Editor,

Hereby I resubmit a new version of the manuscript "Effectiveness of computer-navigated minimally invasive total hip surgery compared to conventional total hip arthroplasty: design of a randomized controlled trial" for publication in the Study Protocol section of the BMC Musculoskeletal Disorders.

The requested formatting changes are made to the manuscript. The section Trial Registration listing the trial registry (Current Controlled Trials) and the unique identifying number (ISRCTN52538512) are added to the abstract.

The names of the authors that were replaced by the term ‘et al.’ are added to the references (references 4, 9, 17, 21, 22, and 28).

Yours sincerely, on behalf of all authors,

Inge Reininga

Department of Orthopaedics
University Medical Center Groningen
Hanzeplein 1 P.O. Box 30.001
9700 RB Groningen
The Netherlands
Telephone : 31-(0)503610549
Fax : 31-(0)50-3611737
E-mail : i.reininga@orth.umcg.nl